Author's response to reviews

Title: An unusual case of metastasis of a pulmonary undifferentiated pleomorphic sarcoma to the right ventricle: a case report

Authors:

Guodong Xu (xuguodong5306750@gmail.com)
Guofeng Shao (guofengshaoliuili@163.com)
Shibao Shi (xinpao@hotmail.com)

Version: 6 Date: 21 April 2013

Author's response to reviews: see over
Dear editors and reviewer:

Thank you for your re-review of our manuscript (Manuscript ID:1844533781878722). We appreciate the concerns and suggestions provided by the reviewer, and have revised our manuscript accordingly. Our point-by-point responses are provided below, and text that has been added or modified from the original text is shown in the revised manuscript with the changes coloured. Upon review of our revised manuscript, we hope that you will find it acceptable for publication in *Journal of Medical Case Reports* and we look forward to your response.

Sincerely yours,

Guodong Xu
Response to the reviewer’s comments:

Reviewer #1:

It would be better to:

1. support this valid statement with a relevant up to date reference

Answer: We appreciate for your valuable comment and feel sorry for our mistakes. By searches of PUBMED, we found that mean survival of the patients whom were treated with surgery is 3 months to 1 year for metastatic cardiac tumors. Relevant content and reference were added to the revised manuscript. We apologize for our mistakes again.

2 limit the statement to once (in the Conclusions) by editing the first occurrence out(two sentences right before the highlighted heading Conclusions in page 9) to avoid repetition and redundancy

Answer: Thank you for your comments. We have edited the first occurrence out to avoid repetition and redundancy. Thank you for your concern again.

3 state a plan for oncologic surveillance (in other words, when is the next follow-up and what investigations they will perform, and how often etc.).

Answer: We appreciate for your valuable comment. The patient will be examined every 3 months by echocardiography in the next 2 years. Relevant contents were added to the revised manuscript with the changes coloured. Thank you for your valuable comments again.
Thank you and the reviewer for the kind advice.

Sincerely yours,

Guodong Xu